Author's response to reviews

**Title:** Nutrition attitudes and knowledge in medical students after completion of an integrated nutrition curriculum compared to a dedicated nutrition curriculum: a quasi-experimental study

**Authors:**

Carolyn O Walsh (carolyn.o.walsh@gmail.com)
Sonja I Ziniel (Sonja.Ziniel@childrens.harvard.edu)
Helen K Delichatsios (HDELICHATSIOS@partners.org)
David S Ludwig (David.Ludwig@childrens.harvard.edu)

**Version:** 2  **Date:** 1 August 2011

**Author's response to reviews:** see over
August 1, 2011

Manuscript 1130743790548056

To the Editorial Staff of BMC Medical Education:

We thank you for the opportunity to revise our manuscript, Nutrition attitudes and knowledge in medical students after completion of an integrated nutrition curriculum compared to a dedicated nutrition curriculum: a quasi-experimental study, for your journal.

We have made the following changes:

**Major revisions**

1. It has been noted that all authors read and approved the final manuscript.

2. Reference 19 has been deleted as its inclusion was in error based on a prior draft.

**Minor revisions**

1. Author qualifications and titles have been removed from the author list

2. The country has been included in all affiliations

3. The heading “Background” has been added. The 'Author Information' section now appears after 'Authors' Contributions', and the 'Ethical Approval' section is after the 'Conclusions'.

4. The only noted highlighting was in the “insert table here” comments, which have been deleted; no tracking was noted. Please let us know if you are seeing something different.

5. 'Insert table/figure here' notes have been deleted

6. Visible vertical lines have been removed from your tables.

Thank you,

Carolyn Walsh